Author's response to reviews

Title:The biobehavioral Women's Health CoOp in Pretoria, South Africa: study protocol for a cluster-randomized design

Authors:

Wendee M Wechsberg (wmw@rti.org)
William A Zule (zule@rti.org)
Jacqueline Ndirangu (jndirangu@rti.org)
Tracy L Kline (tkline@rti.org)
Nathanial F Rodman (nfr@rti.org)
Irene A Doherty (idoherty@rti.org)
Scott P Novak (snovak@rti.org)
Charles van der Horst (charles_vanderhorst@med.unc.edu)

Version:2 Date:30 September 2014

Author's response to reviews:

Dear Proel Hennessy Vargas, Journal Editorial Office
In response to your request, we registered in Clinical Trials.gov December 1, 2011 and was recently re-certified August 2014 as still recruiting.
We will also include both the US and South African IRB approvals and the notice of grant award.
We have added line numbers to the manuscript.
Thank you for this opportunity,
Wendee Wechsberg